Author's response to reviews

Title: Consent to Health Record Linkage: Evidence from a Multi-purpose Longitudinal Survey of a General Population

Authors:

Gundi Knies (gknies@essex.ac.uk)
Jonathan Burton (jburton@essex.ac.uk)
Emanuela Sala (emanuela.sala@unimib.it)

Version: 3 Date: 8 February 2012

Author's response to reviews:

The reviewers did not ask for any revisions to improve the manuscript. Reviewer 2 suggested we include the information leaflet used to assist the collection of informed consent in the appendix. Given the BHPS team provide all study materials online we decided to include a general reference to all study materials at the end of paragraph 1, page 6. Many thanks for pointing out that this would be a welcome addition.